Author’s response to reviews

Title: Limits of the social-benefit motive among high-risk patients: A field experiment on influenza vaccination behaviour

Authors:

Ozan Isler (ozanisler@gmail.com)
Burcu Isler (burcubayrak85@gmail.com)
Orestis Kopsacheilis (orestiskopsacheilis@gmail.com)
Eamonn Ferguson (eamonn.ferguson@nottingham.ac.uk)

Version: 1 Date: 15 Apr 2019

Author’s response to reviews:

We have added all author emails to the title page

We have uploaded the figures as separate files as well as the supplementary files.

Please accept our apologies for having to make these addition changes, The paper was transferred from BMC Medicine.

Please let me know if there any any addition changes needed.

All the best,

Eamonn